Author’s response to reviews

Title: Participatory Learning and Action to address type 2 diabetes in rural Bangladesh: a qualitative process evaluation

Authors:

Joanna Morrison (joanna.morrison@ucl.ac.uk)
Kohenour Akter (mary05kh@gmail.com)
Hannah Maria Jennings (hannah.jennings.11@ucl.ac.uk)
Tasmin Nahar (tasminpcp@dab-bd.org)
Abdul Kuddus (kudduspcp@dab-bd.org)
Sanjit Kumer Shaha (sanjitpcp@dab-bd.org)
Naveed Ahmed (naveedahmed225@gmail.com)
Carina King (c.king@ucl.ac.uk)
Hassan Haghparast-Bidgoli (h.haghparast-bidgoli@ucl.ac.uk)
Anthony Costello (cihdcostello@gmail.com)
A.K. Azad Khan (president@dab-bd.org)
Kishwar Azad (kishwar.azad@googlemail.com)
Edward Fottrell (e.fottrell@ucl.ac.uk)

Version: 4 Date: 04 Oct 2019

Author’s response to reviews:

Dear editors,

My apologies for the mistakes made in the author contribution section.

I have corrected author contributions, which I hope are satisfactory.

Given that we reported on a process evaluation of a large cluster randomised controlled trial, many people were involved in the conception of the study (building on how PLA interventions have worked in the past) and interpretation of the data.
Please let me know if further revisions are required.

Many thanks.